Author's response to reviews

Title: Coding of procedures documented by general practitioners in Swedish primary care - an explorative study using two procedure coding systems

Authors:

Anna Vikström (anna.vikstrom@sll.se)
Maria Hägglund (maria.hagglund@ki.se)
Mikael Nyström (mikael.nystrom@liu.se)
Lars-Erik Strender (lars-erik.strender@ki.se)
Sabine Koch (sabine.koch@ki.se)
Per Hjerpe (per.hjerpe@vgregion.se)
Ulf Lindblad (ulf.lindblad@allmed.gu.se)
Gunnar H Nilsson (gunnar.nilsson@sll.se)

Version: 4 Date: 4 January 2012

Author's response to reviews:

Formatting changes have been made regarding the major revisions, highlighting/tracking, the title of Methods, Authors¿ contributions and typography.